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Live time Nov 27 — Dec 3.
MINERVA 94.7%
MINERVA*MINOS 88.3%

Nov 28 - the DAQ froze.
The DAQ was restarted by
running the DAQ restart
scripts.



v Data

* As described in last weeks AEM talk, down time on Nov 29 was
caused by a power outage due to a sump pump ground fault. This
caused a power failure in the MINOS Hall.

— It took about 4 hours of work before we could get the CROC-Es to
talk to the FEB. After that we quickly returned to data taking.

— In addition the MINOS Hall chiller did not start up after the power
was restored. This caused additional MINOS downtime.



Work Performed During RF
Cavity Replacement

* As reported before every so often the DAQ would crash with a
sequencer error and skip to the next subrun.

— The sequencer is the firmware code in the CROC-E which reads out
the FEBs. This code used to be in the DAQ code. Putting this code in
the CROC-E sped up the DAQ.

— This happens on 2 chains on 2 separate CROC-Es

* For the chain with the most frequent error, we added a short cable to
that chain. This seemed to fix the problem.

— The fix is similar to fixes we did to get the DAQ operating after the
CROCs were replaced by CROC-Es before the start of the ME Run.

— During the next planned one day outage, we will work on the other
chain.

— Christian Gingu is trying to understand the problem.




Rock Muons/POT
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/ Outer CaJorimeter

Event Display

TOWER 4

T

U

View

V-View

[Hit Maps
w . s
% s g 5]
(= | =
o T T T T T T T T T T T T T o
Fizo- X = 20
= 1ewW 3
1004 ¢ - V '—t'loo_
80 80~
604 604
40— 404
204 & 20
o T T T T T T T T T T T T o
w e
P e
& s- % 5
=
= =
[+ B8 2 [+
L 6 ; 1'0 lg 2'0 ZTS 3'0 315 AIID 4E 5‘0 Sg ETO CITS 7'0 715 E’D ETE E 6
s I &
5 5.4 £ s
= =
o LI [ e T T 9%
=1 20-] - X S20-]
= =
100 =100
a0 a0
604 604
40— 40—
204
od—r—r T T T T T
o
g s
=
2
ol ! ]
£ & I43b 15 2b 25 30 35 4b s sh 55 6h 65 70 75 gb &S

G‘WEHJ

. =
g |
=2
T T T T T T T T T T T T T T T T T T T T T
=
i
|
H
T T T T T T T T T T T T
- ]
% 1
= J

15 26 25 30 35 40 a5 5b 55 60 65 70 75 g0 85 9b o51dadnial:

T
1520 253035 4b a5 shs5 6o 65 7075 eb s

Stopping muon with Michel electron in time later slice

6



NuMI| Beam Plots
Dec 1-7

Week ending 00:00 Monday 08 December 2014
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